Author’s response to reviews

Title: Knowledge and practice of adolescent females about menstruation and menstruation hygiene visiting a public healthcare institute of Quetta, Pakistan

Authors:

Judy Michael (judy_michael@hotmail.com)
Qaiser Iqbal (dr.qaiseruob@gmail.com)
Sajjad Haider (drsaajadhaidar007@gmail.com)
Adnan Khalid (dradnankhalid66@gmail.com)
Naheed Haque (afihaque@gmail.com)
Rabia Ishaq (rabiliaishaq72@gmail.com)
Fahad Saleem (fahaduob@gmail.com)
Mohamed Azmi Hassali (azmihassali@gmail.com)
Mohammad Bashaar (dr.mbashaar@gmail.com)

Version: 1 Date: 11 Feb 2019

Author’s response to reviews:

The ethic statement is added to the revised manuscript